Author's response to reviews

Title: A two-way street: bridging implementation science and cultural adaptations of mental health treatments

Authors:

Leopoldo J. Cabassa (ljc2139@columbia.edu)
Ana A. Baumann (abaumann@gwbmail.wustl.edu)

Version: 2 Date: 29 March 2013

Author's response to reviews:

Response to reviewers is within the cover letter that is attached.